Author’s response to reviews

Title: Iatrogenic Hypervitaminosis D as an Unusual Cause of Persistent Vomiting: A Case Report

Authors:

Rinkesh K Bansal (dr.rinkeshbansal@gmail.com)
Pankaj Tyagi (drpankajtyagi@gmail.com)
Praveen Sharma (drpraveen_sharma@yahoo.com)
Vikas Singla (singlavikas1979@gmail.com)
Naresh Bansal (drnbansal@ymail.com)
Ashish Kumar (ashishk10@yahoo.com)
Anil Arora (dranilarora50@gmail.com)

Version: 4 Date: 7 December 2013

Author’s response to reviews: see over
Response to Reviewers Comments

Editor’s Comments:
1. Please change the description of the patient’s gender from 'lady' to ‘female’.
   Response: As advised we have changed the description to ‘female’.

2. Please include a list of abbreviations used in the manuscript and their meanings after the Conclusions section.
   Response: As advised we have now included a list of abbreviations used in the manuscript and their meanings after the Conclusions section.

3. Please include an acknowledgement section at the end of the manuscript before the reference list. Please acknowledge anyone who contributed towards the study by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship. Please also include the source(s) of funding for all authors. Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements.
   Response: As advised we have now included an acknowledgement section and source of funding; however for both these sections the answer is ‘None’.

Reviewer’s Comments:
Reviewer: Indra Ramasamy
Comments:
Please give reference ranges for analytes
Response: As advised we have now included the reference range of all analytes.

P12 refers to mumol/L -?Is this correct
Response: micromol/L.